# Dietary and Pharmacologic Management to Prevent Recurrent Nephrolithiasis in Adults: A Clinical Practice Guideline From the American College of Physicians

Amir Qaseem, MD, PhD; Paul Dallas, MD; Mary Ann Forciea, MD; Melissa Starkey, PhD; and Thomas D. Denberg, MD, PhD, for the Clinical Guidelines Committee of the American College of Physicians\*

**Description:** The American College of Physicians (ACP) developed this guideline to present the evidence and provide clinical recommendations on the comparative effectiveness and safety of preventive dietary and pharmacologic management of recurrent nephrolithiasis in adults.

**Methods:** This guideline is based on published literature on this topic that was identified using MEDLINE, the Cochrane Database of Systematic Reviews (through March 2014), Google Scholar, ClinicalTrials.gov, and Web of Science. Searches were limited to English-language publications. The clinical outcomes evaluated for this guideline include symptomatic stone recurrence, pain, urinary tract obstruction with acute renal impairment, infection, procedure-related illness, emergency department visits, hospitalizations, quality of life, and end-stage renal disease. This guideline grades the quality of evidence and strength of recommendations using ACP's clinical practice guidelines grading system. The target audience for

this guideline is all clinicians, and the target patient population is all adults with recurrent nephrolithiasis ( $\geq 1$  prior kidney stone episode).

**Recommendation 1:** ACP recommends management with increased fluid intake spread throughout the day to achieve at least 2 L of urine per day to prevent recurrent nephrolithiasis. (Grade: weak recommendation, low-quality evidence)

**Recommendation 2:** ACP recommends pharmacologic monotherapy with a thiazide diuretic, citrate, or allopurinol to prevent recurrent nephrolithiasis in patients with active disease in which increased fluid intake fails to reduce the formation of stones. (Grade: weak recommendation, moderate-quality evidence)

Ann Intern Med. 2014;161:659-667. doi:10.7326/M13-2908 www.annals.org For author affiliations, see end of text.

N ephrolithiasis is a condition in which kidney stones, formed from crystals precipitating from the urine, develop within the urinary tract when the urinary concentration of crystal-forming substances is high or that of substances that inhibit stone formation is low. Approximately 80% of adults with kidney stones have stones consisting primarily of calcium oxalate, calcium phosphate, or both. Other stones consist of struvite, uric acid, or cystine. The lifetime prevalence of nephrolithiasis is 13% for men and 7% for women (1, 2), with a 5-year recurrence rate after an initial event of 35% to 50% without treatment (3). Stones are caused by an interaction between genetics and environmental exposure (4).

Efforts to prevent the recurrence of nephrolithiasis target decreasing concentrations of the lithogenic factors (for example, calcium and oxalate) and increasing the concentrations of inhibitors of stone formation (for example, citrate). This is achieved by both dietary changes and appropriate pharmacologic approaches for preventing recurrent kidney stones. Dietary changes include increasing water intake, reducing dietary oxalate, reducing dietary animal protein and other purines, and maintaining normal dietary calcium. The purpose of this American College of Physicians (ACP) guideline is to present the available evidence on the comparative effectiveness and safety of preventive dietary and pharmacologic management for recurrent nephrolithiasis. The target audience for this guideline is all clinicians, and the target patient population is all adults with recurrent nephrolithiasis ( $\geq 1$  prior kidney stone episode). This guideline is based on a systematic evidence review (5) and an evidence report sponsored by the Agency for Healthcare Research and Quality (6).

#### **METHODS**

This guideline is based on a systematic evidence review (5) that addressed the following key questions in adults with a history of nephrolithiasis:

1. Do results of baseline stone composition and blood and urine chemistries predict the effectiveness of diet

#### See also:

Summary for Patients. ..... I-24 Related article: Ann Intern Med. 2013;158:535-43.

<sup>\*</sup> This paper, written by Amir Qaseem, MD, PhD; Paul Dallas, MD; Mary Ann Forciea, MD; Melissa Starkey, PhD; and Thomas D. Denberg, MD, PhD, was developed for the Clinical Guidelines Committee of the American College of Physicians. Individuals who served on the Clinical Guidelines Committee from initiation of the project until its approval were: Thomas D. Denberg, MD, PhD (*Chair*); Michael J. Barry, MD; Molly Cooke, MD; Paul Dallas, MD; Nick Fitterman, MD; Mary Ann Forciea, MD; Russell P. Harris, MD, MPH; Linda L. Humphrey, MD, MPH; Tanveer P. Mir, MD; Holger J. Schünemann, MD, PhD; J. Sanford Schwartz, MD; Paul Shekelle, MD, PhD; and Timothy Wilt, MD, MPH. Approved by the ACP Board of Regents on 23 November 2013.

# *Table 1.* The American College of Physicians' Guideline Grading System\*

Quality of Evidence	Strength of Recommendation			
LVIGENCE	Benefits Clearly Outweigh Risks and Burden or Risks and Burden Clearly Outweigh Benefits	Benefits Finely Balanced With Risks and Burden		
High Moderate Low	Strong Strong Strong	Weak Weak Weak		
	Insufficient evidence to determine n	at hanafita ay viela		

and/or pharmacologic treatment on final health outcomes and intermediate stone outcomes, as well as reduce adverse effects?

2. Do results of follow-up blood and urine biochemistry measurements predict final health outcomes and intermediate stone outcomes in adults being treated to prevent recurrence?

3. What is the effectiveness and comparative effectiveness of different dietary therapies on final health outcomes and intermediate stone outcomes?

4. What is the evidence that dietary therapies to reduce risk for recurrent stone episodes are associated with adverse effects?

5. What is the effectiveness and comparative effectiveness of different pharmacologic therapies on final health outcomes and intermediate stone outcomes?

6. What is the evidence that pharmacologic therapies to reduce risk for recurrent stone episodes are associated with adverse effects?

The literature search included English-language trials identified by using MEDLINE, the Cochrane Database of Systematic Reviews (January 1948 to September 2012), Google Scholar, ClinicalTrials.gov, and Web of Science. The literature search was updated in March 2014, and no additional studies met the inclusion or exclusion criteria. Dietary interventions that were evaluated included intake of fluids, calcium, animal protein, sodium, fruit and fiber, purine, oxalate, potassium, soft drinks, citrus, or multicomponent diets. We also included empirical dietary interventions and those tailored according to patient demographics, comorbid conditions, baseline diet, baseline urine or blood biochemical testing, and/or stone type. Pharmacologic agents evaluated include medications approved by the U.S. Food and Drug Administration and available in the United States for prescription (for example, hydrochlorothiazide, chlorthalidone, indapamide, potassium citrate, potassium-magnesium citrate, sodium citrate, allopurinol, magnesium hydroxide, or acetohydroxamic acid [AHA]). For key questions 1, 2, 4, and 6, we considered final clinical health outcomes as the most important measures of

660 4 November 2014 Annals of Internal Medicine Volume 161 • Number 9

Downloaded From: http://annals.org/ on 12/09/2014

treatment benefit, including symptomatic stone recurrence, pain, urinary tract obstruction with acute renal failure, infection, morbidity related to treatment of a recurrent stone, emergency department visits or hospitalizations for treatment of recurrent stones (for example, for renal colic or acute renal failure), quality of life (general or urologic), and end-stage renal disease. The next most important measures of treatment considered for key questions 1, 2, 4, and 6 were intermediate stone outcomes, including composite recurrence (combination of symptomatic or radiographically detected recurrence), recurrence detected only by scheduled radiographic imaging, and change in stone size. Evidence suggests that stones identified with imaging are associated with symptomatic recurrence. For key questions 3 and 5, adverse effects included any reported by eligible trials (for example, nausea, diarrhea, hypokalemia, weight change, hyperlipidemia, and hyperglycemia). Measures of treatment adherence were those reported by the individual trials (for example, self-report questionnaire, pill count, or as estimated by follow-up urine biochemical measures). Further details about the methods and inclusion and exclusion criteria applied in the evidence review are available in the full Agency for Healthcare Research and Quality report (6).

This guideline rates the quality of evidence and strength of recommendations using ACP's guideline grading system (**Table 1**). Details of the ACP guideline development process can be found in ACP's methods paper (7).

# RELATIONSHIP BETWEEN BASELINE (PRETREATMENT) STONE COMPOSITION AND BIOCHEMISTRY (BLOOD AND URINE) AND TREATMENT EFFICACY TO PREVENT STONE RECURRENCE

Results from 1 good-quality (8) and 28 fair-quality trials (9–35) showed that current evidence is insufficient to conclude that assessing baseline stone composition, blood chemistry, or urine chemistry before initiating pharmacologic or dietary interventions reduces stone recurrence.

# RELATIONSHIP BETWEEN MONITORING (IN-TREATMENT) STONE COMPOSITION AND BIOCHEMISTRY (BLOOD AND URINE) AND TREATMENT EFFICACY TO PREVENT STONE RECURRENCE

Evidence is insufficient from 1 good-quality (8) and 15 fair-quality trials (9, 11–14, 18–21, 27, 28, 30, 32–34) to conclude that monitoring stone composition, blood chemistry, or urine chemistry once pharmacologic or dietary interventions have been initiated reduces stone recurrence.

#### **BENEFITS OF DIETARY THERAPIES**

Eight fair-quality trials were identified that assessed the effectiveness of dietary therapies on recurrent kidney stones

<sup>\*</sup> Adopted from the classification developed by the GRADE (Grading of Recommendations Assessment, Development, and Evaluation) workgroup.

ntervention	Type of Stone Recurrence (Reference)	RR for Stone Recurrence (95% CI)	Quality of Evidence	Adverse Events	RR for Withdrawal (95% C
Dietary intervention					
Increased fluid intake vs. control	Composite (33) Radiographic (29)	0.45 (0.24–0.84) 0.15 (0.02–1.07)	Low Insufficient‡	None reported	1.11 (0.49–2.50)
Increased mineral water intake vs. increased tap water intake	Radiographic (31)	0.73 (0.48–1.09)	Low	None reported	None reported
Reduced soft drink intake vs. control	Symptomatic (35)	0.83 (0.71–0.98); subgroup of soda acidified by phosphoric acid: 0.65 (0.49–0.87)	Low	None reported	1.57 (1.00–2.49)
Decreased animal protein intake vs. control	Composite (28, 30, 32, 34)	1.00 (0.52–1.91)	Low	None reported	0.94 (0.70–1.27)
Increased dietary fiber intake vs. control	Composite (28)	1.18 (0.66–2.12)	Low	None reported	0.89 (0.66–1.21)
Multicomponent diet (high calcium, low protein, and low sodium) vs. control diet	Composite (30)	0.52 (0.29–0.95)	Low	Hypertension, gout, and stroke	0.89 (0.37–2.15); withdrawa due to adverse events, 0.43 (0.12–1.58)
Multicomponent diet (low animal protein and high fiber) vs. control diet	Composite (34)	5.88 (1.39–24.92)	Low	None reported	0.60 (0.29–1.24)
Tailored diet vs. empirical diet	Composite (32)	0.32 (0.14–0.74)	Low	None reported	NA
Pharmacologic intervention					
Thiazide diuretic vs. placebo or control	Symptomatic (14) Composite (9–11, 13, 27)	1.04 (0.39–2.80) 0.53 (0.41–0.68)	Insufficient‡ Moderate	Orthostatic reactions, gastrointestinal upset, erectile dysfunction, fatigue, and muscle symptoms	1.77 (1.12–2.82); withdrawa due to adverse events, 5.15 (1.51–17.57)
Citrate vs. placebo or control	Composite (8, 15, 16, 19)	0.25 (0.14–0.44)	Moderate	Gastrointestinal symptoms	1.76 (1.12–2.75); withdrawa due to adverse events, 5.18 (1.51–17.79)
Allopurinol vs. placebo or	Radiographic (18) Symptomatic (20)	0.95 (0.62–1.44) 0.36 (0.11–1.19)	Low	Rash, acute gout, and leukopenia	0.81 (0.41–1.62); withdrawa
control	Composite (20, 23) Radiographic (20)	0.59 (0.42–0.84) 1.07 (0.16–7.10)	Moderate Insufficient‡	Rasii, acute gout, and ieuropenia	due to adverse events, 0.53 (0.16–1.77)
AHA vs. placebo or control	Radiographic (24, 25)	0.81 (0.18–3.66)	Insufficient‡	Anemia (RR, 2.18 [CI, 1.13–4.21]), headache (RR, 2.18 [CI, 0.42–11.32]), alopecia (RR, 4.36 [CI, 0.51–37.53]), tremor, and deep venous thrombosis	1.31 (0.56–3.05); withdrawa due to adverse events, 4.39 (2.02–9.55)
Magnesium vs. placebo	Composite (27)	0.65 (0.37–1.16)	Low	Diarrhea	1.09 (0.40–2.97); withdrawa due to adverse events, 1.82 (0.20–16.77)
Thiazide diuretic plus citrate vs. thiazide diuretic	Composite (9)	0.94 (0.52–1.68)	Low	None reported	None reported
Thiazide diuretic plus allopurinol vs. thiazide diuretic	Composite (11)	0.79 (0.18–3.49)	Insufficient‡	None reported for combination group; both hypokalemia and hypotension reported in 1 participant receiving thiazide monotherapy	0.17 (0.02–1.29); withdrawa due to adverse events, 0.20 (0.01–3.97)

Table 2. Evidence for Prevention of Stone Recurrence With Dietary and Pharmacologic Interventions\*

AHA = acetohydroxamic acid; NA = not applicable; RR = relative risk.

\* Adapted from reference 6.

+ Overall quality graded as insufficient because of the small number of recurrent stone events and imprecise risk estimates.

(28–35). Most studies assessed treatment in patients with calcium stones. Low-quality evidence showed that increased fluid intake; reduced soft drink intake (particularly soda acidified by phosphoric acid); and a high-calcium, low-protein, low-sodium multicomponent diet compared with a control diet reduced stone recurrence. Further, a tailored diet compared with an empirical diet reduced

stone recurrence. A summary of the evidence is presented in Table 2.

#### Increased Fluid Intake Versus No Treatment

Low-quality evidence from 1 study (33) showed that patients with calcium stones who increased fluid intake to achieve more than 2 L of urine per day had less composite

4 November 2014 Annals of Internal Medicine Volume 161 • Number 9 661

# CLINICAL GUIDELINE | Dietary and Pharmacologic Management to Prevent Recurrent Nephrolithiasis

stone recurrence than the control group (12.1% vs. 27.0%; follow-up, 60 months). Another trial (29) of radiographic stone recurrence in patients with calcium stones showed a nonstatistically significant decrease in stone recurrence in patients with increased fluid intake compared with no treatment (8.0% vs. 55.6%; follow-up, 24 to 36 months).

# Increased Mineral Water Intake Versus Increased Tap Water Intake

Oligomineral water is a type of mineral water often marketed for diuretic properties. Low-quality evidence from 1 study (31) of radiographic stone recurrence showed that patients with calcium stones who were assigned to drink more than 2 L of a particular brand of mineral water each day (Fiuggi brand oligomineral water) that contained 15 mg of calcium per liter versus tap water with 55 to 130 mg of calcium per liter showed that 17.0% had recurring stones compared with 22.9%, respectively. No studies assessed the effect of other types of mineral or oligomineral waters on stone recurrence.

#### Decreased Soft Drink Intake Versus No Intervention

Low-quality evidence from 1 study (35) showed that patients (any stone type) with baseline soft drink consumption of more than 160 mL per day who were instructed to abstain from drinking soda had a reduced risk for symptomatic stone recurrence compared with no treatment (33.7% vs. 40.6%). Subgroup analysis showed that the benefit was limited to patients who drank soda that was acidified by phosphoric acid (typically colas) rather than those acidified by citric acid (typically fruit-flavored sodas) (29.7% vs. 45.6%).

#### Multicomponent Dietary Interventions Versus Control

Three RCTs (30, 32, 34) reported mixed results in comparing multicomponent diets with control diets in patients with calcium stones. One fair-quality study (34) showed that 24.0% of patients on a multicomponent diet (low animal protein, high fiber, increased bran, and low purine) had composite stone recurrence compared with 4.1% of those on a control diet. Patients on the multicomponent diet also had a higher rate of composite stone recurrence (7.1 vs. 1.2 per 100 person-years). One fairquality trial (30) showed that 20.0% of patients on a diet that included normal to high calcium, low animal protein, and low sodium had composite stone recurrence compared with 38.3% of patients on a low-calcium control diet. A third fair-quality study (32) showed that fewer patients who received an extensive metabolic evaluation and subsequent dietary modification had increased composite stone recurrence than patients who received a limited biochemical evaluation with general dietary recommendations (6.2% vs. 19.1%).

#### High Fiber Intake Versus Usual Diet

Low-quality evidence from 1 study (28) of patients with calcium stones showed no statistically significant difference between composite stone recurrence for those on a

662 4 November 2014 Annals of Internal Medicine Volume 161 • Number 9

high-fiber diet (increase baseline fiber intake to 25 g per day through increased fruit, fiber, and whole grain consumption) compared with patients on their usual diet. Two other trials (32, 34) of patients assigned to high fiber intake as part of a multicomponent diet compared with a control diet reported conflicting results.

#### Low Animal Protein Intake Versus Control Diet

Low-quality evidence from 1 study (28) showed no difference between a diet low in animal protein (<3 servings of meat or fish per week and <100 g of milk products per day) and a control diet for risk for composite recurrent stones (47.8% for both groups) in patients with calcium stones. Three other trials included low animal protein as part of a multicomponent diet compared with a control diet; 2 of these found a lower risk for recurrent stones (30, 32), whereas the other (34) found a higher risk for recurrent stones with the multicomponent diet.

#### **Other Dietary Interventions**

One trial (30) in patients with calcium oxalate stones found that fewer patients treated with a multicomponent diet including normal or high calcium intake (1200 mg/d) had recurrent stones than those on a low-calcium diet (400 mg/d); further, patients on a multicomponent diet including low sodium intake (50 mmol/d) were less likely to experience stone recurrence than those on the control diet. Multicomponent diets including low purine intake were assessed in 2 trials (32, 34) with mixed results. No included studies addressed a low-oxalate diet.

### HARMS OF DIETARY THERAPIES

Evidence showed that withdrawals from studies for any cause were low because of increased fluid intake. However, withdrawals were high in long-term trials evaluating soft drink intake, high fiber intake, low animal protein intake, and multicomponent dietary interventions. Most trials had poor reporting of adverse events.

#### Increased Fluid Intake Versus No Treatment

Average withdrawals from the 2 trials (29, 33) were 9.5% (range 0% to 10.0%) and were similar between increased fluid intake and no treatment. One trial (29) reported no withdrawals due to adverse events.

#### Increased Mineral Water Intake Versus Increased Tap Water Intake

One study (31) reported no withdrawals from either group, and they did not report adverse events.

#### Decreased Soft Drink Intake Versus No Intervention

One study (35) reported that 8.7% of patients instructed to abstain from soft drinks withdrew compared with 5.5% of patients given no treatment (relative risk, 1.57 [95% CI, 1.00 to 2.49]).

#### Multicomponent Dietary Interventions Versus Control

No increased risk for withdrawals for multicomponent dietary interventions compared with control was found,

and the average from 3 trials was 16.4%. One trial reported that 5.0% of withdrawals were due to adverse events for the multicomponent diet compared with 11.7% for the control diet, including hypertension (2.0% vs. 12.0%; relative risk, 0.14 [CI, 0.02 to 1.13]), stroke (2.0% vs. 0%), and gout (2.0% vs. 0%) (30). Another trial reported an 18% withdrawal rate for the multicomponent diet compared with 29% for the control diet and no adverse events were reported (34).

#### High Fiber Intake Versus Control Diet

One study (28) reported no increase in withdrawals for patients on the high-fiber diet compared with the control diet (55.0% vs. 61.7%).

#### Low Animal Protein Intake Versus Control Diet

One study (28) reported no increase in withdrawals for a low-protein diet compared with a control diet (58.2% vs. 61.7%).

#### BENEFITS OF PHARMACOLOGIC THERAPIES

A total of 20 (1 good-quality and 19 fair-quality) trials (8-27) assessed pharmacologic therapy for reducing stones. Both control and therapy groups were instructed to increase fluid intake. Most studies assessed treatment in patients with calcium stones. Thiazide, citrate, and allopurinol treatment decreased stone recurrence.

#### Monotherapy

#### Thiazide Diuretic Monotherapy Versus Placebo or Control

Moderate-quality evidence from 6 fair-quality trials compared thiazide diuretic with placebo (13, 14, 27) or control (9–11) in patients with recurrent calcium stones. The results showed that the risk for composite stone recurrence was lower in persons treated with thiazide than placebo or control (24.9% vs. 48.5%) (9–13, 27). No significant differences were found in the risk for recurrence between the different types of thiazides studied or varying dosages.

#### Citrate Monotherapy Versus Placebo or Control

Moderate-quality evidence was derived from 6 trials comparing citrate monotherapy with placebo (8, 19) or control (15–18) in patients with nephrolithiasis (calcium stones in 5 of 6 studies; stone type not specified for 1 study). Pooled data from 4 trials (8, 15, 16, 19) showed that composite stone recurrence was lower in patients treated with citrate than placebo or control (11.1% vs. 52.3%). Results were similar for different types of citrate used, including potassium citrate (16, 19), potassium– magnesium citrate (8), and potassium–sodium citrate (15). One fair-quality trial (18) showed no difference between citrate and control for risk for radiographic stone recurrence.

www.annals.org

#### Allopurinol Monotherapy Versus Placebo or Control

Moderate-quality evidence from 4 trials that compared allopurinol monotherapy with placebo (20, 21) or control (22, 23) in patients with calcium oxalate stones showed a reduced risk for composite stone recurrence with allopurinol. Two trials (20, 23) reporting on composite stone recurrence showed that risk for recurrence was lower in patients treated with allopurinol than placebo (33.3% vs. 55.4%). One fair-quality study (20) showed a reduction in symptomatic stone recurrence (10.3% vs. 29.0%), although they found no difference in recurrence of radiographic stones between the treatments.

#### AHA Monotherapy Versus Control

Evidence from 2 fair-quality trials showed no statistically significant difference in symptomatic or radiographic stone recurrence between AHA treatment compared with placebo or control group (24, 25) in patients with struvite stones. The overall quality of evidence for this outcome was graded as insufficient because of the small number of recurrent stone events and imprecise risk estimates.

#### Magnesium Monotherapy Versus Placebo or Thiazide Diuretic

Low-quality evidence from 1 fair-quality study (27) showed that risk for composite stone recurrence in patients with calcium stones treated with magnesium was nonstatistically significantly lower than placebo (29.4% vs. 45.2%) and nonstatistically significantly higher than thiazide diuretic (29.4% vs. 14.3%).

#### **Combination Therapy**

#### Thiazide Diuretic Plus Citrate Combination Therapy Versus Thiazide Diuretic Monotherapy

Evidence from 1 trial (9) showed no difference in risk for composite stone recurrence between treatment with thiazide plus citrate compared with thiazide monotherapy (30.0% vs. 32.0%). The overall quality of evidence for this outcome was graded as insufficient because of the small number of recurrent stone events and imprecise risk estimates.

#### Thiazide Diuretic Plus Allopurinol Combination Therapy Versus Thiazide Diuretic Monotherapy

Low-quality evidence from 1 study (11) showed that the risk for composite stone recurrence was similar in patients with recurrent calcium oxalate treated with thiazide plus allopurinol compared with thiazide monotherapy (12.5% vs. 15.8%).

#### HARMS OF PHARMACOLOGIC THERAPIES

Evidence related to harms of pharmacologic therapies were derived largely from trials reporting on withdrawals from the studies or withdrawals due to adverse events, and trials were variable. Specific adverse event reporting was poor.

4 November 2014 Annals of Internal Medicine Volume 161 • Number 9 663

### CLINICAL GUIDELINE | Dietary and Pharmacologic Management to Prevent Recurrent Nephrolithiasis

#### Monotherapy

#### Thiazide Diuretic Monotherapy Versus Placebo or Control

Pooled data from 7 fair-quality trials (9–14, 27) showed an increase in withdrawals for thiazides compared with placebo or control (17.0% vs. 8.0%; relative risk, 1.77 [CI, 1.12 to 2.82]), and more withdrawals due to adverse events were in the thiazide treatment group (8.0% vs. 1.0%). Four trials (10, 13, 14, 27) reported adverse events as a composite outcome, including orthostasis, gastrointestinal upset, erectile dysfunction, fatigue, and muscle symptoms, although the list of adverse events was different for each study. One of two trials reporting risk for composite adverse events according to treatment group found an increase with thiazide treatment (27, 34). No adverse events were reported for more than 1 patient in any of the other trials.

#### Citrate Monotherapy Versus Placebo or Control

Data from 4 trials (1 good and 3 fair quality) showed an increased risk for withdrawal for citrate therapy compared with placebo or control (36.0% vs. 20.0%) (8, 17– 19) and an increased risk for withdrawals due to adverse events (15.0% vs. 2.0%).

#### Allopurinol Monotherapy Versus Placebo or Control

Two fair-quality trials showed no increased risk for withdrawals (31.0% vs. 42.0%) and no increased risk for withdrawals due to adverse events (4.0% vs. 8.0%) when comparing allopurinol with placebo (20, 23).

#### AHA Monotherapy Versus Placebo

Two fair-quality trials reported withdrawal rates for 63% for patients receiving AHA therapy versus 46% for placebo (24, 25) and an increased risk for withdrawals due to adverse events for AHA compared with placebo (22.0% vs. 5.0%). An increased risk for any adverse event with AHA versus placebo (64.0% vs. 32.0%) was found, including anemia (18.0% vs. 8.0%) (24, 25), headache (9.0% vs. 4.0%) (25), and alopecia (9.0% vs. 2.0%) (25). Results for tremor (25.0%) and deep venous thrombosis (16.0%) each were reported only for the AHA group in 1 trial (26).

#### Magnesium Monotherapy Versus Placebo

One fair-quality trial (27) found no increased risk for withdrawal for magnesium compared with placebo (18.0% vs. 17.0%). Rates of withdrawal due to adverse events were 6.0% for the magnesium group and 3.0% for the placebo group.

#### **Combination Therapy**

#### Thiazide Diuretic Plus Citrate Combination Therapy Versus Thiazide Diuretic Monotherapy

One fair-quality trial reported no withdrawals for either the thiazide plus citrate group or the thiazide treat-

664 4 November 2014 Annals of Internal Medicine Volume 161 • Number 9

ment groups, and this study did not report adverse events (9).

#### Thiazide Diuretic Plus Allopurinol Combination Therapy Versus Thiazide Diuretic Monotherapy

One fair-quality trial (11) reported withdrawal rates of 4.0% in the thiazide plus allopurinol group, 24.0% in the thiazide-only group, and 16.0% in the control group. Withdrawals due to adverse events were reported in 8% of patients in the thiazide group and none for thiazide plus allopurinol or control groups.

#### **SUMMARY**

Increased fluid intake was shown to decrease stone recurrence by at least half with no reported side effects. Decreasing soft drink intake in men with a high baseline intake of soft drinks acidified by phosphoric acid also decreased stone recurrence. Other dietary interventions, including low-protein diets and multicomponent interventions, showed mixed results. Pharmacologic therapies plus increased fluid intake was effective, and thiazide, citrate, and allopurinol treatment resulted in a statistically significant decrease in stone recurrence. No trials were identified that directly compared these treatments with each other. Clinical outcomes and adverse events were sparsely reported, but they were more common for pharmacologic than nonpharmacologic therapies. Evidence was insufficient to determine the effect of dietary or pharmacologic therapy based on stone composition or blood and urine chemistries. Most studies included only patients with calcium stones, and no trials assessed treatment in patients with uric acid or cystine stones. See the Figure for a summary of the recommendations and clinical considerations.

#### **R**ECOMMENDATIONS

Recommendation 1: ACP recommends management with increased fluid intake spread throughout the day to achieve at least 2 L of urine per day to prevent recurrent nephrolithiasis. (Grade: weak recommendation, low-quality evidence)

Low-quality evidence showed that increased fluid intake is associated with a reduction in stone recurrence. Evidence also did not show any difference between tap water and a specific brand of mineral water (Fiuggi brand oligomineral water). People who already drink recommended amounts of liquids and those in whom increased fluid intake is contraindicated should not be directed to increase their fluid intake further. Although some lowquality evidence shows that a decrease in the consumption of soft drinks is associated with a reduced risk for stone recurrence, this benefit was limited to patients who drank soft drinks acidified by phosphoric acid, such as colas, but not for drinks acidified by citric acid, such as fruit-flavored sodas. *Figure.* Summary of the American College of Physicians guideline on dietary and pharmacologic management to prevent recurrent nephrolithiasis in adults.



### SUMMARY OF THE AMERICAN COLLEGE OF PHYSICIANS GUIDELINE ON DIETARY AND PHARMACOLOGIC MANAGEMENT TO PREVENT RECURRENT NEPHROLITHIASIS IN ADULTS

Disease/Condition	Recurrent nephrolithiasis		
Target Audience	Internists, family physicians, and other clinicians		
Target Patient Population	Adults with recurrent nephrolithiasis		
Interventions Evaluated	Dietary: increased fluid intake, increased oligomineral water intake, decreased soft drink intake, multicomponent dietary interventions, high fiber intake, and low animal protein intake Pharmacologic: thiazide diuretic, citrate, allopurinol, acetohydroxamic acid, and magnesium		
Outcomes Evaluated	Symptomatic stone recurrence, radiographic and composite stone recurrence, pain, urinary tract obstruction with acute renal impairment, infection, procedure-related morbidity, emergency department visits, hospitalizations, quality of life, and ESRD		
Benefits	Decreased stone recurrence		
Harms	Adverse events associated with dietary interventions:		
	Multicomponent diet: hypertension, gout, and stroke		
	Adverse events associated with pharmacologic interventions:		
	Thiazides: orthostasis, gastrointestinal upset, erectile dysfunction, fatigue, and muscle symptoms		
	Citrates: gastrointestinal symptoms		
	Acetohydroxamic acid: anemia, headache, alopecia, tremor, and deep venous thrombosis		
	Allopurinol: rash, acute gout, and leukopenia		
	Magnesium: diarrhea		
Recommendations	Recommendation 1: ACP recommends management with increased fluid intake spread throughout the day to achieve at least 2 L of urine per day to prevent recurrent nephrolithiasis. (Grade: weak recommendation, low-quality evidence)		
	Recommendation 2: ACP recommends pharmacologic monotherapy with a thiazide diuretic, citrate, or allopurinol to prevent recurrent nephrolithiasis in patients with active disease in which increased fluid intake fails to reduce the formation of stones. (Grade: weak recommendation, moderate-quality evidence)		
Inconclusive Areas of Evidence	Relationship between pretreatment or in-treatment stone composition and biochemistry (blood and urine) with treatment efficacy to prevent stone recurrence		
Clinical Considerations	Evidence is applicable primarily to calcium stones.		
	Evidence showed that patients who decreased intake of soda that was acidified by phosphoric acid had reduced kidney stone recurrence. Clinicians should encourage patients to avoid colas as opposed to fruit-flavored soft drinks, which are often acidified by citric acid.		

ESRD = end-stage renal disease.

Recommendation 2: ACP recommends pharmacologic monotherapy with a thiazide diuretic, citrate, or allopurinol to prevent recurrent nephrolithiasis in patients with active disease in which increased fluid intake fails to reduce the formation of stones. (Grade: weak recommendation, moderatequality evidence)

Moderate-quality evidence showed that thiazide diuretics, citrates, and allopurinol reduce the risk for recurrence of composite calcium stones. Combination therapy with these agents was not more beneficial than monotherapy. Although biochemistry and some observational data on stone recurrence suggest that the choice of treatment could be based on the type of metabolic abnormality, evidence from randomized, controlled trials is lacking to correlate the drug of choice and stone type to the prevention of stone recurrence. Most patients have calcium stones, and evidence showed that thiazide diuretics, citrates, and allopurinol all effectively reduced recurrence of this stone type. Note that the available evidence evaluated

www.annals.org

Downloaded From: http://annals.org/ on 12/09/2014

higher doses of thiazides (hydrochlorothiazide, 50 mg; chlorthalidone, 25 or 50 mg; indapamide, 2.5 mg) to prevent recurrent nephrolithiasis. The use of lower doses of thiazides is associated with fewer adverse effects, but their effectiveness in preventing stone recurrence compared with higher doses is not known. All of the medications are associated with adverse events, which are summarized in Table 2.

#### **INCONCLUSIVE AREAS OF EVIDENCE**

The current evidence from randomized trials is insufficient to evaluate the benefits of knowing the stone composition, urine chemistry, and blood chemistry related to the effectiveness of treatment. The guidelines committee is aware that observational data show an association of stone composition and biochemistry with stone recurrence; moreover, physiologic knowledge suggests that interventions targeting stone composition, biochemistry, or both

4 November 2014 Annals of Internal Medicine Volume 161 • Number 9 665

# CLINICAL GUIDELINE | Dietary and Pharmacologic Management to Prevent Recurrent Nephrolithiasis

can favorably alter biochemical composition that leads to stone formation. Clinicians often select pharmacologic therapy on the basis of method of action: Thiazide diuretics reduce urinary calcium and are often prescribed for patients with hypercalciuria, citrates bind to calcium and decrease urine acidity, and allopurinol decreases uric acid in urine. Almost all studies analyzed in the evidence review included only patients with calcium stones, which are the most common stone type. Although biochemistry suggests a relationship between pharmacologic method of action and stone type, no randomized, controlled trials link biochemical testing to outcomes. Our recommendations do not include patients with suspected hyperparathyroidism or other rare cases.

From the American College of Physicians and University of Pennsylvania Health System, Philadelphia, Pennsylvania, and Virginia Tech Carilion School of Medicine and Carilion Clinic, Roanoke, Virginia.

**Note:** Clinical practice guidelines are "guides" only and may not apply to all patients and all clinical situations. Thus, they are not intended to override clinicians' judgment. All ACP clinical practice guidelines are considered automatically withdrawn or invalid 5 years after publication, or once an update has been issued.

**Disclaimer:** The authors of this article are responsible for its contents, including any clinical or treatment recommendations.

Financial Support: Financial support for the development of this guideline comes exclusively from the ACP operating budget.

**Disclosures:** Authors followed the policy regarding conflicts of interest described at www.annals.org/article.aspx?articleid=745942. Disclosures can be viewed at www.acponline.org/authors/icmje/ConflictOfInterest Forms.do?msNum=M13-2908. Any financial and nonfinancial conflicts of interest of the group members were declared, discussed, and resolved. A record of conflicts of interest is kept for each Clinical Guidelines Committee meeting and conference call and can be viewed at www .acponline.org/clinical\_information/guidelines/guidelines/conflicts\_cgc .htm.

Requests for Single Reprints: Amir Qaseem, MD, PhD, MHA, American College of Physicians, 190. N. Independence Mall West, Philadelphia, PA 19106; e-mail, aqaseem@acponline.org.

Current author addresses and author contributions are available at www.annals.org.

#### References

1. Stamatelou KK, Francis ME, Jones CA, Nyberg LM, Curhan GC. Time trends in reported prevalence of kidney stones in the United States: 1976-1994. Kidney Int. 2003;63:1817-23. [PMID: 12675858]

2. Pearle MS, Calhoun EA, Curhan GC; Urologic Diseases of America Project. Urologic diseases in America project: urolithiasis. J Urol. 2005;173:848-57. [PMID: 15711292]

3. Uribarri J, Oh MS, Carroll HJ. The first kidney stone. Ann Intern Med. 1989;111:1006-9. [PMID: 2688503]

4. Attanasio M. The genetic components of idiopathic nephrolithiasis. Pediatr Nephrol. 2011;26:337-46. [PMID: 20563734] doi:10.1007/s00467-010 -1562-6 5. Fink HA, Wilt TJ, Eidman KE, Garimella PS, MacDonald R, Rutks IR, et al. Medical management to prevent recurrent nephrolithiasis in adults: a systematic review for an American College of Physicians Clinical Guideline. Ann Intern Med. 2013;158:535-43. [PMID: 23546565] doi:10.7326/0003-4819 -158-7-201304020-00005

6. Fink HA, Wilt TJ, Eidman KE, Garimella PS, MacDonald R, Rutks IR, et al. Recurrent Nephrolithiasis in Adults: Comparative Effectiveness of Preventive Medical Strategies. Comparative Effectiveness Review No. 61. (Prepared by the University of Minnesota Evidence-based Practice Center under Contract No. 290-02-0009.) AHRQ Publication No. 12-EHC049-EF. Rockville, MD: Agency for Healthcare Research and Quality; 2012.

7. Qaseem A, Snow V, Owens DK, Shekelle P; Clinical Guidelines Committee of the American College of Physicians. The development of clinical practice guidelines and guidance statements of the American College of Physicians: summary of methods. Ann Intern Med. 2010;153:194-9. [PMID: 20679562] doi: 10.7326/0003-4819-153-3-201008030-00010

8. Ettinger B, Pak CY, Citron JT, Thomas C, Adams-Huet B, Vangessel A. Potassium-magnesium citrate is an effective prophylaxis against recurrent calcium oxalate nephrolithiasis. J Urol. 1997;158:2069-73. [PMID: 9366314]

9. Fernández-Rodríguez A, Arrabal-Martín M, García-Ruiz MJ, Arrabal-Polo MA, Pichardo-Pichardo S, Zuluaga-Gómez A. [The role of thiazides in the prophylaxis of recurrent calcium lithiasis]. Actas Urol Esp. 2006;30:305-9. [PMID: 16749588]

10. Ahlstrand C, Sandwall K, Tiselius HG. Prophylactic treatment of calcium stone formers with hydrochlorothiazide and magnesium. In: HG Tiselius, ed. Renal stones — Aspects on Their Formation, Removal and Prevention. Proceedings of the Sixth European Symposium on Urolithiasis 1995. Edsbruk: Akademitryck; 1995:195-7.

11. Borghi L, Meschi T, Guerra A, Novarini A. Randomized prospective study of a nonthiazide diuretic, indapamide, in preventing calcium stone recurrences. J Cardiovasc Pharmacol. 1993;22 Suppl 6:S78-86. [PMID: 7508066]

12. Ala-Opas M, Elomaa I, Porkka L, Alfthan O. Unprocessed bran and intermittent thiazide therapy in prevention of recurrent urinary calcium stones. Scand J Urol Nephrol. 1987;21:311-4. [PMID: 2832935]

 Laerum E, Larsen S. Thiazide prophylaxis of urolithiasis. A double-blind study in general practice. Acta Med Scand. 1984;215:383-9. [PMID: 6375276]
Scholz D, Schwille PO, Sigel A. Double-blind study with thiazide in recurrent calcium lithiasis. J Urol. 1982;128:903-7. [PMID: 7176047]

15. Lojanapiwat B, Tanthanuch M, Pripathanont C, Ratchanon S, Srinualnad S, Taweemonkongsap T, et al. Alkaline citrate reduces stone recurrence and regrowth after shockwave lithotripsy and percutaneous nephrolithotomy. Int Braz J Urol. 2011;37:611-6. [PMID: 22099273]

16. Soygür T, Akbay A, Küpeli S. Effect of potassium citrate therapy on stone recurrence and residual fragments after shockwave lithotripsy in lower caliceal calcium oxalate urolithiasis: a randomized controlled trial. J Endourol. 2002;16: 149-52. [PMID: 12028622]

17. Premgamone A, Sriboonlue P, Disatapornjaroen W, Maskasem S, Sinsupan N, Apinives C. A long-term study on the efficacy of a herbal plant, Orthosiphon grandiflorus, and sodium potassium citrate in renal calculi treatment. Southeast Asian J Trop Med Public Health. 2001;32:654-60. [PMID: 11944733]

18. Hofbauer J, Höbarth K, Szabo N, Marberger M. Alkali citrate prophylaxis in idiopathic recurrent calcium oxalate urolithiasis—a prospective randomized study. Br J Urol. 1994;73:362-5. [PMID: 8199822]

19. Barcelo P, Wuhl O, Servitge E, Rousaud A, Pak CY. Randomized doubleblind study of potassium citrate in idiopathic hypocitraturic calcium nephrolithiasis. J Urol. 1993;150:1761-4. [PMID: 8230497]

20. Ettinger B, Tang A, Citron JT, Livermore B, Williams T. Randomized trial of allopurinol in the prevention of calcium oxalate calculi. N Engl J Med. 1986; 315:1386-9. [PMID: 3534570]

21. Miano L, Petta S, Galatioto GP, Gullucci M. A placebo controlled doubleblind study of allopurinol in severe recurrent idiopathic renal lithiasis. In: Schwille PO, Smith LH, Robertson WG, Vahlensieck W, eds. Urolithiasis and Related Clinical Research. New York: Plenum Pr; 1985:521-4.

22. Robertson WG, Peacock M, Sepby PL, Williams RE, Clark P, Chisholm GD, et al. A multicentre trial to evaluate 3 treatments for recurrent idiopathic calcium stone disease—a preliminary report. New York: Plenum Pr; 1985:545-48.

23. Smith MJ. Placebo versus allopurinol for renal calculi. J Urol. 1977;117: 690-2. [PMID: 875139]

666 4 November 2014 Annals of Internal Medicine Volume 161 • Number 9

#### Dietary and Pharmacologic Management to Prevent Recurrent Nephrolithiasis | CLINICAL GUIDELINE

24. Griffith DP, Khonsari F, Skurnick JH, James KE. A randomized trial of acetohydroxamic acid for the treatment and prevention of infection-induced urinary stones in spinal cord injury patients. J Urol. 1988;140:318-24. [PMID: 3294442]

25. Griffith DP, Gleeson MJ, Lee H, Longuet R, Deman E, Earle N. Randomized, double-blind trial of Lithostat (acetohydroxamic acid) in the palliative treatment of infection-induced urinary calculi. Eur Urol. 1991;20:243-7. [PMID: 1726639]

26. Williams JJ, Rodman JS, Peterson CM. A randomized double-blind study of acetohydroxamic acid in struvite nephrolithiasis. N Engl J Med. 1984;311:760-4. [PMID: 6472365]

27. Ettinger B, Citron JT, Livermore B, Dolman LI. Chlorthalidone reduces calcium oxalate calculous recurrence but magnesium hydroxide does not. J Urol. 1988;139:679-84. [PMID: 3280829]

28. Dussol B, Iovanna C, Rotily M, Morange S, Leonetti F, Dupuy P, et al. A randomized trial of low-animal-protein or high-fiber diets for secondary prevention of calcium nephrolithiasis. Nephron Clin Pract. 2008;110:c185-94. [PMID: 18957869] doi:10.1159/000167271

29. Sarica K, Inal Y, Erturhan S, Yagci F. The effect of calcium channel blockers on stone regrowth and recurrence after shock wave lithotripsy. Urol Res. 2006; 34:184-9. [PMID: 16463053]

30. Borghi L, Schianchi T, Meschi T, Guerra A, Allegri F, Maggiore U, et al. Comparison of two diets for the prevention of recurrent stones in idiopathic hypercalciuria. N Engl J Med. 2002;346:77-84. [PMID: 11784873]

31. Di Silverio F, Ricciuti GP, D'Angelo AR, Fraioli A, Simeoni G. Stone recurrence after lithotripsy in patients with recurrent idiopathic calcium urolithiasis: efficacy of treatment with fiuggi water. Eur Urol. 2000;37:145-8. [PMID: 10705191]

32. Kocvara R, Plasgura P, Petrík A, Louzenský G, Bartonícková K, Dvorácek J. A prospective study of nonmedical prophylaxis after a first kidney stone. BJU Int. 1999;84:393-8. [PMID: 10468751]

33. Borghi L, Meschi T, Amato F, Briganti A, Novarini A, Giannini A. Urinary volume, water and recurrences in idiopathic calcium nephrolithiasis: a 5-year randomized prospective study. J Urol. 1996;155:839-43. [PMID: 8583588]

34. Hiatt RA, Ettinger B, Caan B, Quesenberry CP Jr, Duncan D, Citron JT. Randomized controlled trial of a low animal protein, high fiber diet in the prevention of recurrent calcium oxalate kidney stones. Am J Epidemiol. 1996;144: 25-33. [PMID: 8659482]

35. Shuster J, Jenkins A, Logan C, Barnett T, Riehle R, Zackson D, et al. Soft drink consumption and urinary stone recurrence: a randomized prevention trial. J Clin Epidemiol. 1992;45:911-6. [PMID: 1624973]



# **Annals of Internal Medicine**

Current Author Addresses: Drs. Qaseem and Starkey: 190 N. Independence Mall West, Philadelphia, PA 19106.Dr. Dallas: Virginia Tech Carilion School of Medicine, 1906 Bellview Avenue, Roanoke, VA 24014.

Dr. Forciea: University of Pennsylvania Health System, 3615 Chestnut Street, Philadelphia, PA 19104.

Dr. Denberg: Carilion Clinic, PO Box 13727, Roanoke, VA 24036.

Author Contributions: Conception and design: A. Qaseem, M.A. Forciea, T. Denberg.

Analysis and interpretation of the data: A. Qaseem, P. Dallas, M.A. Forciea, M. Starkey, T. Denberg.

Drafting of the article: A. Qaseem, P. Dallas, M.A. Forciea, M. Starkey, T. Denberg.

Critical revision of the article for important intellectual content: A. Qaseem, P. Dallas, M.A. Forciea, M. Starkey, T. Denberg.

Final approval of the article: A. Qaseem, P. Dallas.

Statistical expertise: A. Qaseem.

Administrative, technical, or logistic support: A. Qaseem, M. Starkey. Collection and assembly of data: A. Qaseem, M. Starkey.